Author’s response to reviews

Title: Effectiveness of a multidisciplinary care program on recovery and return to work of patients after gynaecological surgery; design of a randomized controlled trial

Authors:

Antonie Vonk Noordegraaf (t.vonknoordegraaf@vumc.nl)
Judith A.F. Huirne (j.huirne@vumc.nl)
Hans A.M. Brölmann (h.brolmann@vumc.nl)
Mark H. Emanuel (memanuel@spaarneziekenhuis.nl)
Paul J.M. van Kesteren (p.j.m.vankesteren@olvg.nl)
Gunilla Kleiverda (gkleiverda@flevoziekenhuis.nl)
Jos P. Lips (jlips@kg.nl)
Alexander Mozes (almo@zha.nl)
Andreas Thurkow (a.thurkow@slaz.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Johannes R. Anema (h.anema@vumc.nl)

Version: 3 Date: 19 January 2012

Author’s response to reviews: see over
Dear Sir, Madam,

As requested, a point-by-point description of the changes made in this revised version, using your questions regarding requested changes to the format of our manuscript:

**Major revisions (we require the author to make these changes)**

1. **Author list:** the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ‘First name Initial Last name’ (e.g. Joe F Bloggs) => performed

2. **Affiliations:** Please provide the city for affiliation 10 => performed

3. **Ethics:** Please place the ethics section in the Methods/design section

4. **Tables:** The images in table 2 cross over cells, if you wish to have images in cells they must not cross table lines => I have adapted table 2

5. **Additional files:** Please remove the additional files => performed

**Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)**

6. **Keywords:** Please remove the keywords from the submission system copy of the abstract => performed

7. **Tables:** Please place the tables after the references => performed

8. **Tables:** Please ensure all table lines are black => performed

9. **Tables:** please remove the visible vertical lines from your tables => performed
10. Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order => I’m not completely sure whether I do understand this question correctly, because in the previous version of this manuscript, I already made a paragraph with abbreviations, which are spelt out and in alphabetical order. So I didn’t change this section, but if there is still anything you want me to change, please let me know.

11. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. => performed

Kind regards,

Ton Vonk Noordegraaf
Research physician

VU University Medical Center
Department of Public and Occupational Health Van der Boechorststraat 7
NL-1081 BT Amsterdam
The Netherlands
T:  +31 20 444 6065
M:  +31 61 808 8761
F:  +31 20 444 8387
E-mail: t.vonknoordegraaf@vumc.nl